



**Baishideng  
Publishing  
Group**

7901 Stoneridge Drive, Suite 501,  
Pleasanton, CA 94588, USA  
**Telephone:** +1-925-223-8242  
**Fax:** +1-925-223-8243  
**E-mail:** [bpgoffice@wjgnet.com](mailto:bpgoffice@wjgnet.com)  
**https://**[www.wjgnet.com](http://www.wjgnet.com)

Dear Editor;

**Manuscript NO:** 44859

**Title:** Intraperitoneal Rupture of the Hydatid Cyst: Four Case Report and Literature Review

**Reviewer's code:** 02528812

**Reviewer Comment:** The manuscript entitled "Intraperitoneal Rupture of the Hydatid Cyst: Four Case Report and Literature Review" is well presented. In this study, the authors have presented four cases of spontaneous rupture of cyst hydatid in patients aged between 27 and 44. Three of the perforated cysts were in liver and the other one was located in spleen. These cases were successfully treated by albendazole after surgical intervention. This is an informative and valuable study. As intraperitoneal hydatid cyst rupture is a life threatening complication and is often misdiagnosed, this study is very useful to both clinicians and patients. I have no specific comments to the authors, only some minor language polishing should be checked.

**Response to the Reviewer:** Thank you for your positive comment related with our manuscript



**Baishideng  
Publishing  
Group**

7901 Stoneridge Drive, Suite 501,  
Pleasanton, CA 94588, USA  
**Telephone:** +1-925-223-8242  
**Fax:** +1-925-223-8243  
**E-mail:** bpgoffice@wjgnet.com  
**https://**www.wjgnet.com

**Reviewer's code:** 02942798

**Reviewer Comment**

Dear sir, thank you to select me to review manuscript Akbulut S, Ozdemir F et al. Intraperitoneal Rupture of the Hydatid Cyst: Four Case Report and Literature Review. Authors described 4 patients with intraperitoneal rupture of the hydatid cyst, 2 cases were spontaneous and 2 were posttraumatic. Paper is well written, but it needs a great deal of language polishing. Case 2: exophytic, not egzofitic Case 3: diagnosed, not dignosed Discussion: Budd Chiari, not budd - chiari (>10 cm), not (>10) macroscopis, not macroskobic literature, not literatüre

**Response to the Reviewer:** Thank you for your positive comment related with our manuscript. Necessary corrections were made according to your suggestions.

Best Regards

**Correspondence to:** Sami Akbulut, Assos Prof, FICS, FACS, Department of Surgery and Liver Transplant Institute, Inonu University Faculty of Medicine, Elazig Yolu 10. Km, Malatya 44280, Turkey. sami.akbulut@inonu.edu.tr, akbulutsami@gmail.com

Telephone: +90-422-3410660

Fax: +90-422-3410036